Specific comments: Title: Natural isothiocyanates of the genus Capparis as potential agonists of apoptosis and antitumor drugs by Hanuš et al. Cancer is a severe and progressively advancing disease. In this context, the study by Hanuš et al. sheds light on the field of Phytomedicine based on medicinal plants in cancer. These authors have focused on isothiocyanates, which exhibit active and significant anticancer activity, along with their anticancer compounds. In this manner, these authors have endeavored to establish a scientific foundation for the traditional use of this medicinal plant in cancer treatment. #I have observed that the topic exhibits originality and relevance within the field, effectively addressing a specific gap in current research. #The methodology is fine and no further control is required. #I found the conclusion to be in line with the evidence and arguments presented. #The references are well-updated. #The Figures and Tables are ok. #Only minor English language editing is required. Overall Nice Work!

Answering Reviewers: The authors have made corrections to the text and completely improved the English.